**Author’s response to reviews**

**Title:** Factors promoting Iranian older adults’ spirituality: A qualitative content analysis

**Authors:**

Fatemeh Hajinejad (faty278@yahoo.com)

Ebrahim Ebrahimi (evarash@yahoo.com)

Anneke de Jong (anneke.dejong@hu.nl)

Maryam Ravanipour (ravanipour@gmail.com)

**Version:** 1 Date: 30 Mar 2018

**Author’s response to reviews:**

Dear Editor-in-Chief of BMC Geriatrics,

I am writing to resubmit our manuscript entitled, “Factors promoting Iranian older adults’ spirituality: A qualitative content analysis” for exclusive consideration for publication as an article in BMC Geriatrics.

A point-by-point responses to the valued comments of the two dear reviewers have been added in a table at the end of this letter. We ensure that all changes to the manuscript are indicated in the text by yellow highlighting.

This paper was approved and financially sponsored by the Deputy of Research of the Bushehr University of Medical Sciences (Code:22892).

This manuscript is not under consideration by any other journal and all authors approved the manuscript and this submission. It would be your kindness if you check the eligibility of our manuscript; hope we have this chance to publish our paper in your distinguished journal.

Yours sincerely,

Maryam Ravanipour,

Associate Professor of Nursing,
Bushehr University of Medical Sciences,
Bushehr, Iran